Author's response to reviews

Title: Institutional delivery service utilization and associated factors among mothers who gave birth in the last 12 months in Sekela District, North West of Ethiopia: A community - based cross sectional study Alemayehu Shimeka, Fekadu Mazengia and Solomon Meseret

Authors:

Alemayehu S Shimeka (alemayehushimeka@gmail.com)
Fekadu M Mazengia (mazengiafek23@gmail.com)
Solomon M Woldeyohannes (messislo@yahoo.com)

Version: 2 Date: 10 May 2012

Author's response to reviews: see over
Dear Sir/Madam;

Thanks once again for considering our manuscript important for BMC Public Health. We have enclosed revised manuscript entitled as "Institutional delivery service utilization and associated factors among mothers who gave birth in the last 12 months in Sekela District, North West of Ethiopia" to be considered for publication in BMC Pregnancy and Childbirth. In this revised manuscript, we have attempted to address all of the reviewers’ comments. As most of the comments are forwarded from reviewer two we have uploaded separate response letter to Reviewer 2. We have also accommodated the comment from the reviewer one (Luwei Pearson) but we have not included separate response letter.

Respectfully,

Alemayehu Shimeka (MPH in Epidemiology)